RESEARCH





Prevalence and predictors of premarital sexual intercourse among young women in sub-Saharan Africa

Eugene Budu^{1,2}, Abdul-Aziz Seidu^{3,4,5}, Ebenezer Kwesi Armah-Ansah¹, James Boadu Frimpong^{6,7*}, Richard Gyan Aboagye⁸, Stephen Kofi Anin^{9,10}, John Elvis Hagan Jr^{6,11} and Bright Opoku Ahinkorah^{3,12}

Abstract

Introduction Premarital sexual intercourse (PSI) without adequate information and/or appropriate application of the relevant knowledge about sex before marriage, potentially has adverse effects on the sexual and reproductive health outcomes of vulnerable young women in sub-Saharan Africa (SSA). This study sought to examine the prevalence and predictors of PSI among young women aged 15–24 in SSA.

Methods Nationally representative cross-sectional data from 29 countries in SSA were extracted for the study. A weighted sample size of 87,924 never married young women was used to estimate the prevalence of PSI in each country. A multilevel binary logistic regression modelling approach was used to examine the predictors of PSI at p < 0.05.

Results The prevalence of PSI among young women in SSA was 39.4%. Young women aged 20–24 (aOR=4.49, 95% CI=4.34, 4.65) and those who had secondary/higher educational level (aOR=1.63, 95% CI=1.54, 1.72) were more likely to engage in PSI compared to those aged 15–19 and those with no formal education. However, young women who belonged to the Islamic religion (aOR=0.66, 95% CI=0.56, 0.78); those who were working (aOR=0.75, 95% CI=0.73, 0.78); belonged to the richest wealth index (aOR=0.55, 95% CI=0.52, 0.58); were not exposed to radio at all (aOR=0.90, 95% CI=0.81, 0.99); were not exposed to television at all (aOR=0.50, 95% CI=0.46, 0.53); resided in rural areas (aOR=0.73, 95% CI=0.70, 0.76); and those who were living in the East African sub-region (aOR=0.32, 95% CI=0.29, 0.35) were less likely to engage in PSI compared to those who were traditionalist, unemployed, belonged to the poorest wealth index, exposed to radio frequently, exposed to television frequently, resided in urban areas, and lived in the Southern Africa sub-region, respectively.

Conclusion Sub-regional variations in the prevalence of PSI exist amidst multiple risk factors among young women in SSA. Concerted efforts are required to empower young women financially, including education on sexual and reproductive health behaviors such as the detrimental effects of sexual experimentation and encouraging abstinence and/or condom use through regular youth-risk communication advocacy.

*Correspondence: James Boadu Frimpong frimpongboadujames@gmail.com Full list of author information is available at the end of the article



© The Author(s) 2023. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

Plain Language Summary

Having premarital sexual intercourse (PSI) without adequate knowledge and application of the knowledge could have adverse effects on the sexual and reproductive health of vulnerable young women in sub-Saharan Africa (SSA). This study examined the prevalence and predictors of PSI among young women in SSA. Nationally representative cross-sectional data from 29 countries in SSA were used. A sample size of 87,924 never married young women was used to estimate the prevalence of PSI. A multilevel binary logistic regression was used to examine the predictors of PSI. The prevalence of PSI among young women in SSA was high. Young women aged 20–24 and those who had attained secondary/higher educational level were more likely to engage in PSI. However, young women who belonged to the Islamic religion; were working; belonged to the richest wealth index; were not exposed to radio at all; were not exposed to television at all; resided in rural areas; and those who were living in the East African sub-region were less likely to engage in PSI. Sub-regional variations in the prevalence of PSI exist amidst multiple risk factors among young women in SSA. Concerted efforts are required to empower young women financially, including education on sexual and reproductive health behaviors such as the detrimental effects of sexual experimentation and encouraging abstinence and/or condom use through regular youth-risk communication advocacy.

Keywords Premarital sexual intercourse, Sexual and reproductive health, sub-Saharan Africa, Young women

Introduction

The world has seen a significant increase in the total number of young people and it has been estimated that more than three-quarters of the world's 1.8 billion young people reside in developing regions including sub-Saharan Africa (SSA) [1, 2]. Majority of these young people engage in premarital sexual intercourse (PSI) or sex before they attain their second decade in life [1].

PSI among young women has become a major public health concern as the world began witnessing a significant increase in reported sexual activities leading to teenage pregnancies and abortions in most developing regions [2–6]. PSI has been explained to encompass penetrative vaginal intercourse that occurs between two individuals before he/she starts a formal marriage life [5, 7–9]. PSI in SSA has been defined as sexual intercourse before attaining age 18 [10, 11]. This is the more vulnerable group of people who are mostly not married and are young [12].

Although PSI is common among young people, it does not translate into contraceptive use [13]. This has had an adverse effect on their sexual and reproductive health outcomes [14, 15]. Young people are involved in PSI without adequate information and knowledge of reproductive health and contraceptives [9, 16–18]. It is essential to note that many life events, health-damaging behaviours, and PSI start at these youthful years [2]. Therefore, the onset of sexual intercourse is a potentially life-changing milestone in the physical and psychological development of women in all settings and the timing and circumstances within which it happens can have either an immediate or long-term consequences on their health and wellbeing [19].

At this stage of development, young women are faced with exploration and risk-taking behaviours [20]. These risky behaviours are associated with pressure to use alcohol, cigarettes, drugs, early sexual relationships, intentional and unintentional injuries, having multiple sexual partners, violence that could lead to unwanted pregnancy, unsafe abortion, and sexually transmitted infections (STIs) including HIV/AIDS, hasty and unpromising marriage, lesser employment opportunity, unplanned parenthood, and dropping out of school [2, 6, 7, 21].

It has been documented that young women in SSA tend to engage in early sexual debut than their male counterparts and it is one of the major predictors that put young women at a high-risk of HIV/AIDS [11, 22]. This is because majority of the young women are challenged with peer pressure and are shy to discuss sexual matters with their parents explicitly due to culture reasons [1].

Studies have revealed that there are three categories of predictors of PSI among young women in SSA: namely individual, family, and institutional level predictors [12, 22]. The individual-level predictors include age, sex, ethnicity, love affair, and loneliness [12]. The family-level predictors consist of family type, family income, occupation, broken families, and parenting while the institutional level predictors are not only limited to social network, organization, communication (mobile phones, internet, books and magazines, radio and television) but includes policies and laws [12, 23]. Also, studies have indicated that factors such as urban poor settings, high unemployment, unstable wages, low literacy, and inadequate recreational facilities have been linked to young women's engagement in PSI and multiple sexual partners [14, 24, 25].

There is a high prevalence of risky sexual behaviours (e.g., multiple sexual partners, transactional sex, early sexual debut) among young women in SSA [26, 27]. Evidence suggest that by age 15 years, young women in Kenya would have had at least a sexual intercourse in their lifetime while the median age at first sexual intercourse among young women in Ghana and Malawi is 16 years [22].

Studies on sexual practices and behaviours in SSA have mainly focused on adolescents [1]; correlates of early sexual debut [22]; university students [28, 29]; preparatory school students living with and without parents [18]; magnitude and associated factors of premarital sex [2]; in-school youths [16]; parenting style [6]; impact of premarital sex on health [12]; and adolescent students' attitude towards premarital sex [23].

Since there is a paucity of studies related to the prevalence and predictors of premarital sexual intercourse among young women in SSA, it is essential to understand and identify current trend of this behavioral practice of young women in order to develop appropriate and context-specific interventions in the sub-region. Findings from this study will play an invaluable role in inspiring young women about the risk of premarital sexual practice and inform policymakers as well as for organizations that work in this area.

Methods

Data source and study design

Data for the study were extracted from the most recent Demographic and Health Surveys (DHS) of 29 countries in SSA. We pooled the data from the women's recode files in each of the countries. The DHS is a comparatively nationally representative survey conducted in over 85 low-and-middle-income countries worldwide [30]. DHS employs a descriptive cross-sectional design. Respondents for the survey were recruited using a two-stage cluster sampling method. Detailed sampling technique has been highlighted in the literature [31]. Standardized structured questionnaires were used to collect data from the respondents on health and social indicators including age at first sexual intercourse. We included a total of 87,924 never married young women in our study. Only the women with complete cases of variables of interest were included in the study using listwise deletion. The dataset used is freely available at https://dhsprogram. com/data/available-datasets.cfm. This manuscript was drafted with reference to the Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) statement guidelines [32]. Table 1 provides details of the description of the study sample.

 Table 1
 Description of the study sample

Country	Survey year	Weighted sample (n)	Weighted percentage (%)
1. Angola	2015-16	3974	4.5
2. Burkina Faso	2010	2754	3.1
3. Benin	2017-18	3556	4.0
4. Burundi	2016-17	5071	5.8
5. Congo DR	2013-14	4139	4.7
6. Congo	2011-12	2287	2.6
7. Cote d'Ivoire	2011-12	2294	2.6
8. Cameroon	2018	3650	4.1
9. Ethiopia	2016	3448	3.9
10. Gabon	2012	2284	2.6
11. Ghana	2014	2407	2.7
12. Gambia	2019–20	3095	3.5
13. Guinea	2018	2366	2.7
14. Kenya	2014	3352	3.8
15. Comoros	2012	1426	1.6
16. Liberia	2019-20	2169	2.5
17. Lesotho	2014	359	0.4
18. Mali	2018	1495	1.7
19. Malawi	2015-16	4768	5.4
20. Nigeria	2018	3716	4.2
21. Namibia	2013	2856	3.2
22. Rwanda	2019-2020	3999	4.5
23. Sierra Leone	2019	4080	4.6
24. Senegal	2010-11	3738	4.2
25. Chad	2014-15	2599	3.0
26. Togo	2013-14	2164	2.7
27. Uganda	2016	4165	4.7
28. Zambia	2018	3545	4.0
29. Zimbabwe	2015	2165	2.5
All countries	2010-2020	87,924	100.0

Variables

Outcome variable

The study used PSI as the outcome variable. PSI refers to any sexual relations a person has prior to marriage [33]. Simply put, it is when a person engages in sexual activities before they marry. Restricting the analytical sample to never married young women, this variable was derived using the question on age at first sex which was "at what age did [NAME] first have sex?" For this study, those who had never had sex were put in the category "No PSI" and this category was coded as "0" while those who had sex at age eight (which was the minimum age at first sex for the study sample in all the countries) and above were put in the category "2. Studies that used the DHS dataset employed similar coding [34–36].

Explanatory variables

The explanatory variables considered in this study were selected based on their association with PSI from literature [21, 28, 37] and also their availability in the DHS dataset. A total of 10 variables were included in the study. These variables were grouped into individual level and contextual level factors. The individual level factors considered in this study were women's age, educational level, religious affiliation, occupational status, wealth index, frequency of reading newspaper or magazine, frequency of listening to radio, and frequency of watching television. Place of residence and geographical subregion were the contextual level variables in the study. The categories of each of the variables are shown in Table 2.

Statistical analyses

Data for the study were analysed using Stata version 16. First, a bar chart was used to show the prevalence of PSI among young women across the 29 countries. Next, the weighted frequencies and percentages for the explanatory variables were presented as indicated in Table 2. Later, we presented the bivariate results on the distribution of PSI among young women across the explanatory variables using the Pearson chi-square test of independence (Table 2). We further conducted a cross-tabulation between age and the other explanatory variables as a sensitivity test to show how age is distributed across the other explanatory variables (Additional file 1). After this, we checked for collinearity multicollinearity among the explanatory variables using the variance inflation factor (VIF) and the results showed no evidence of high collinearity (Maximum VIF = 1.66, Minimum VIF = 1.07and Mean VIF = 1.31). Finally, a four-model multilevel binary logistic regression (Model O-III) analysis was conducted. Model O was an empty model where no explanatory variable was used. Model I had only the individual level variables. Model II had only the contextual variables while Model III, which was considered as the complete model had both the individual and contextual level variables. The results were presented as adjusted odds ratio (aOR) with their respective 95% confidence interval (CI). All the analyses were weighted while the survey command (svy) in Stata was used to adjust for the complex sampling structure of the data in the analyses.

Ethical consideration

In this study, ethical clearance was not sought due to the public availability of the DHS dataset. The datasets were obtained from the Monitoring and Evaluation to Assess and Use Results Demographic and Health Surveys (MEASURE DHS) after registration and approval were given for its usage. All the ethical guidelines concerning the use of secondary datasets in the publication were strictly adhered to. Detailed information about the DHS data usage and ethical standards are available at http:// goo.gl/ny8T6X.

Results

Prevalence of premarital sexual intercourse among young women in SSA

Figure 1 displays the prevalence of PSI among young women in SSA. The prevalence of PSI among young women in SSA was 39.4%. The country-specific prevalence indicated that Liberia recorded the highest (75.3%) prevalence whereas Comoros recorded the lowest (5.0%) prevalence.

Association between premarital sexual intercourse and explanatory variables

Table 2 outlines the results on the association between PSI and the explanatory variables. The study found that age, level of education, religion, employment status, wealth index, exposure to newspaper or magazine, exposure to radio, exposure to television, place of residence, and sub-region were all significantly associated with PSI at p < 0.001.

Predictors of premarital sexual intercourse among young women in sub-Saharan Africa

Table 3 presents the results of the predictors of PSI among young women in SSA. The study found that young women aged 20-24 (aOR=4.49, 95% CI=4.34, 4.65) and those who had secondary/higher educational level (aOR=1.63, 95% CI=1.54, 1.72) had higher odds of engaging in PSI compared to those aged 15-19 and those with no formal education. However, young women who belonged to the Islamic religion (aOR = 0.66, 95%) CI = 0.56, 0.78; were working (aOR = 0.75, 95% CI = 0.73, 0.78); belonged to the richest wealth index (aOR = 0.55, 95% CI=0.52, 0.58); were not exposed to radio at all (aOR=0.90, 95% CI=0.81, 0.99); were not exposed to television at all (aOR=0.50, 95% CI=0.46, 0.53); resided in rural areas (aOR=0.73, 95% CI=0.70, 0.76); and those who were living in the East Africa sub region (aOR = 0.32, 95% CI=0.29, 0.35) had lower odds of engaging in PSI compared to those who were traditionalist, not working, belonged to the poorest wealth index, exposed to radio almost every day, exposed to television almost every day, resided in urban areas and living in the Southern Africa sub-region, respectively.

Table 2 Distribution c	of premarital	sexual intercourse	among young	y women ir	n sub-Saharan	Africa across the	explanatory	variables
(n=87,924)								

Variables	Frequency	Percentage	Premarital sexual intercourse	P-value
Age				< 0.001
15–19	62,329	70.9	29.0	
20–24	25,595	29.1	64.6	
Level of education				< 0.001
No formal education	9648	11.0	28.6	
Primary	25,576	29.1	33.0	
Secondary/higher	52,700	59.9	44.4	
Religion				< 0.001
Christianity	61,719	70.2	44.4	
Islam	24,255	27.6	25.8	
Traditional	692	0.8	30.9	
No religion	1258	1.4	56.7	
Employment status				< 0.001
Not working	49,005	55.7	35.7	
Working	38,919	44.3	44.0	
Wealth index				< 0.001
Poorest	10,756	12.2	36.2	
Poorer	13,520	15.4	37.2	
Middle	16,244	18.5	40.0	
Richer	20,020	22.8	41.2	
Richest	27,384	31.1	39.9	
Frequency of reading newspaper or	magazine			< 0.001
Not at all	61,699	70.2	37.9	
Less than once a week	13,766	15.7	40.2	
At least once a week	11,539	13.1	44.4	
Almost every day	920	1.0	58.1	
Frequency of listening to radio				< 0.001
Not at all	35,100	39.9	38.2	
Less than once a week	18,424	21.0	38.7	
At least once a week	31,582	35.9	39.2	
Almost every day	2818	3.2	59.0	
Frequency of watching television				< 0.001
Not at all	39,641	45.2	35.7	
Less than once a week	13,054	14.8	39.2	
At least once a week	27,993	31.8	39.4	
Almost every day	7235	8.2	59.6	
Place of residence				< 0.001
Urban	43,434	49.4	45.2	
Rural	44,490	50.6	33.7	
Sub-region	,			< 0.001
West Africa	31,665	36.0	38.1	
East Africa	37.874	43.1	32.9	
Central Africa	15,169	17.2	53.1	
Southern Africa	3216	3.7	63.6	

Average

Liberia

39.35





Fig. 1 Prevalence (%) of premarital sexual intercourse among young women in sub-Saharan Africa

 Table 3
 Multilevel binary logistic regression results on the predictors of premarital sexual intercourse among young women in sub-Saharan Africa

Variables	Null model	Model I aOR (95% CI)	Model II aOR (95% CI)	Model III aOR (95% CI)
Age				
15-19		Reference (1.0)		Reference (1.0)
20-24		4.39*** (4.25-4.55)		4.49*** (4.34-4.65)
Level of education				
No formal education		Reference (1.0)		Reference (1.0)
Primary		1.09** (1.03-1.15)		1.34*** (1.26–1.42)
Secondary/higher		1.49*** (1.41-1.58)		1.63*** (1.54–1.72)
Religion				
Christianity		1.49*** (1.26-1.75)		1.74*** (1.47-2.06)
Islam		0.67*** (0.56-0.79)		0.66*** (0.56-0.78)
Traditional		Reference (1.0)		Reference (1.0)
No religion		2.44*** (1.99-2.98)		2.44*** (1.99-3.00)
Employment status				
Not working		0.77*** (0.75–0.80)		0.75*** (0.73–0.78)
Working		Reference (1.0)		Reference (1.0)
Wealth index				
Poorest		Reference (1.0)		Reference (1.0)
Poorer		0.99 (0.94-1.04)		0.99 (0.94–1.05)
Middle		0.96 (0.91–1.01)		0.94* (0.89–0.99)
Richer		0.83*** (0.78–0.87)		0.77*** (0.73–0.82)
Richest		0.59*** (056-0.62)		0.55*** (0.52-0.58)
Frequency of reading nev	wspaper or magazine	(000 0.02)		0.00 (0.02 0.00)
Not at all	ispaper of magazine	1 06 (0 89–1 25)		0.98 (0.83–1.17)
Less than once a week		0.89 (0.75–1.05)		0.86 (0.72–1.02)
At least once a week		0.98 (0.83-1.17)		0.87 (0.73–1.04)
Almost every day		Beference (1.0)		Beference (1.0)
Frequency of listening to	radio	hereitete (1.0)		herefelee (1.0)
Not at all	lucio	0.82*** (0.74–0.91)		0.90* (0.81_0.99)
Loss than onco a wook		0.02 (0.74 0.91)		0.00 (0.01 0.00)
At losst opco a wook		0.60*** (0.62, 0.76)		0.00 (0.72-0.03)
Almost overv day		0.09 (0.02-0.70) Reference (1.0)		0.79 (0.71-0.00) Reference (1.0)
Frequency of watching to	Novision	Reference (1.0)		Reference (1.0)
Not at all				
Not at all		0.42 (0.39-0.43)		0.50 (0.40-0.53)
Less than once a week		0.51 (0.47-0.55)		0.50 (0.52-0.00)
Al reast once a week		0.55 (0.51 - 0.59)		0.50 (0.52 - 0.00)
Almost every day		Reference (1.0)		Reference (1.0)
Place of residence			Deference (10)	Deference (10)
Dural			Reference (1.0)	0.72*** (0.70, 0.76)
Rulai			0.74 (0.72-0.70)	0.75 (0.70-0.70)
VVest Africa			0.34**** (0.32=0.37)	0.01**** (0.50-0.07)
Edst AIrica			0.28**** (0.20-0.31)	0.32**** (0.29-0.35)
Central Alfica			$0.05^{}(0.00-0.71)$	0.78 (0.71–0.86)
Southern Africa			Kelerence (1.0)	reference (1.0)
Random effect results		0.04 (0.04, 0.05)		
PSU variance	0.0126451	0.04 (0.04-0.06)	0.0141602	0.0127214
	0.0126461	0.0132099	0.0141608	0.013/314
LR test	250.07 (X = 0.0000)	214.16 (X = 0.0000)	262./4 (X = 0.0000)	203.97 (X = 0.0000)

Variables	Null model	Model I aOR (95% CI)	Model II aOR (95% CI)	Model III aOR (95% CI)
Wald chi-square		11,593.86***	2934.43***	13,307.89***
Model fitness				
Log-likelihood	- 58,809.755	- 51,886.136	- 57,288.897	- 50,442.408
AIC	117,623.5	103,816.3	114,589.8	100,936.8
Ν	87,924	87,924	87,924	87,924
Number of groups	1567	1567	1567	1567

aOR adjusted odds ratio, CI confidence interval

*p<0.05, **p<0.01, ***p<0.001

Discussion

The study examined the prevalence and predictors of PSI among young women in SSA. The study found that the pooled prevalence of PSI among young women in SSA was 39.4%. The country-specific prevalence indicated that Liberia recorded the highest (75.3%) prevalence whereas Comoros recorded the lowest (5.0%). The variations in the prevalence of PSI among young women noted in this study could be attributed to the differences in sociocultural practices in these countries. For example, it could be that young women in Comoros normalized premarital sexual behaviours or certain societal norms encouraged sexual experimentation among young women [1], which made them underreport their engagement in PSI. The fear that they would be identified and punished could have also led to the lower prevalence of PSI among young women from Comoros. The highest prevalence of PSI among young women in Liberia could be as a result of the young women's incessant engagement in transactional sex [38, 39]. Our finding indicates that young women in SSA could be at higher risk of contracting STIs. Therefore, both public and private organizations should intensify their efforts to educate young women about the consequences of engaging in PSI.

Similar to the findings of other previous studies [1, 16, 20, 40], our study found that the likelihood of PSI among young women heightened with increasing age. A plausible explanation for this finding could be as result of older women engaging in romantic relationships, which increase their likelihood of having PSI [16, 41, 42]. Since younger women aged 15–19 are generally expected to be in school, it is not surprising that they are less likely to have PSI. This finding suggests that sexual and reproductive health education should be targeted at older young women to reduce the occurrence of PSI in SSA.

Akin to the observation of previous study [1], this study found that young women who had secondary/ higher educational level had higher odds of engaging in PSI compared to those with no formal education. This finding could be as a result of the influence of peers who convince their colleagues to engage in erotic relationships, which subsequently increases their likelihood of engaging in premarital sex [43, 44]. It could also be that young women who have attained some level of education try to experiment sexual activities that might have been discussed in the schools, increasing their likelihood of having PSI [45, 46]. This finding implies that having higher education is positively associated with PSI, hence, more concerted efforts should be put in educating young women in the school environment about the detrimental effects of sexual experimentation while encouraging abstinence or contraceptive use.

However, young women who belonged to the Islamic religion had lower odds of engaging in PSI relative to those who were traditionalists. This finding is similar to a study [47] which found that being more religious decreased the likelihood of sexual debut among females. Our finding could be attributed to the existence of certain Islamic teachings that frown upon PSI, reducing young women's likelihood to engage in sexual activities before marriage [47, 48]. Another reason for this finding could be that engaging in more religious activities such as praying five times daily and Quranic recitals protect young Islamic women from engaging in PSI [47].

Young women who were working had lower odds of engaging in PSI compared to those who were not working. Our finding could be that women who are employed are less encouraged to engage in PSI due to the workload which pose as a source of stress on them. Also, women who are unemployed may have the desire to engage in PSI because of curiosity, experimentation and for financial gains to sustain themselves economically [49, 50]. This finding signifies that providing young women with economically sustainable jobs and empowering them financially could help reduce the occurrence of PSI in SSA. Contrary to the finding of a previous study [51], this study revealed that young women who belonged to the richest wealth index had lower odds of engaging in PSI as against those who belonged to the poorest wealth index. This finding was observed probably because being financially buoyant protects people from engaging in risky sexual behaviors such as streetism, substance use, and transactional sex which increase women's propensity of having PSI [52].

Despite the well-documented positive influence of mass media exposure (e.g., television and radio) on sexual behavior [26, 53], this study revealed that young women who were not exposed to radio and television at all had lower odds of engaging in PSI compared to those who were exposed to radio and television almost every day. A probable explanation for this unexpected finding could be attributed to the influx of excess sexually explicit telenovelas and advertisement of sexual enhancing drugs that increase the desire of young women to have PSI [54]. For PSI among young women in SSA to reduce significantly, it is important for media outlets to regulate the programs they churn out for public consumption.

Contrary to the finding of previous studies [1, 55], it was found in this study that young women who resided in rural areas had lower odds of engaging in PSI compared to those who resided in urban areas. A plausible explanation for this finding could be linked with the relatively lower standard of living in rural areas, which reduces young women's likelihood of engaging in transactional sex for financial assistance [56]. Even though this is not clear, this finding calls for further studies to substantiate this probable reason.

Young women who were living in the East Africa subregion had lower odds of engaging in PSI compared to those who were living in the Southern Africa sub-region. A plausible reason for this finding could be attributed to the variations in the socio-cultural practices among the sub-regions of SSA. For example, it is perceived in some parts of Southern Africa sub-region that having a child increases the chance of marriage, which also increases the rate of PSI among young women [57]. This was evident in this study where Southern African countries such as Angola, Lesotho, Namibia, and Zambia recorded a high prevalence of premarital sex among young women.

Strengths and limitations

Since the subject was very sensitive, ruling out social desirability bias in relation to the responses was impossible. Additionally, the cross-sectional nature of the study makes it difficult to establish causal inferences among the studied variables. Again, since the study used self-reported questionnaires, it could be that respondents might have underreported or over-reported their experiences. Despite the above-mentioned limitations, the study has certain strengths that need to be mentioned. Hence, findings from the study should be interpreted with caution. First, the study employed a relatively large sample size that is nationally representative and would be appropriate for generalization. Moreover, the use of statistical procedures to generate interesting findings that could be verified is a strength for the study.

Conclusions

The study revealed that the prevalence of PSI among young women in SSA was high with Liberia and Comoros recording the highest (75.3%) and lowest (5.0%) prevalences, respectively. The study has also identified factors that predict PSI among young women in the studied countries, hence, interventions that seek to alleviate PSI among young women should pay critical attention to these factors. The study recommends that more concerted efforts should be directed at empowering young women financially and educating them about their sexual and reproductive health behaviors such as the detrimental effects of sexual experimentation and also encouraging abstinence and/or contraceptive use.

Abbreviations

SSA	Sub-Saharan Africa
PSI	Premarital sexual intercourse
DHS	Demographic and Health Surveys
aOR	Adjusted odds ratio
CI	Confidence interval
STIs	Sexually transmitted illnesses
STROBE	Strengthening the Reporting of Observational Studies in
	Epidemiology
VIF	Variance inflation factor

Supplementary Information

The online version contains supplementary material available at https://doi. org/10.1186/s12978-023-01626-8.

Additional file 1: Table S1. Distribution of age across explanatory variables.

Acknowledgements

We acknowledge MEASURE DHS for making the data freely available for the study.

Author contributions

Study conception: AS, BOA, JEH. Data analysis: EB, RGA, AS, BOA. Writing of manuscript: EB, EKA-A, JBF, AS, RGA, SKA, JEH, BOA. All authors read and approved the final manuscript.

Funding

No specific funding was obtained for this study.

Availability of data and materials

The dataset used for the study is freely available at https://dhsprogram.com/ data/available-datasets.cfm.

Declarations

Ethics approval and consent to participate

In this study, ethical clearance was not sought due to the public availability of the DHS dataset. The datasets were obtained from the MEASURE DHS after registration and approval were given for its usage. All the ethical guidelines concerning the use of secondary datasets in the publication were strictly adhered to. Detailed information about the DHS data usage and ethical standards are available athttp://goo.gl/ny8T6X.

Consent for publication

Not applicable.

Competing interests

The authors declare that they have no competing interests.

Author details

¹Department of Population and Health, University of Cape Coast, Cape Coast, Ghana.²Korle Bu Teaching Hospital, P.O.Box 77, Accra, Ghana.³REMS Consult, Takoradi, Ghana. ⁴Centre for Gender and Advocacy, Takoradi Technical University, P.O. Box 256, Takoradi, Ghana. ⁵College of Public Health, Medical and Veterinary Sciences, James Cook University, Townsville, Australia. ⁶Department of Health, Physical Education, and Recreation, University of Cape Coast, Cape Coast, Ghana. ⁷Department of Kinesiology, New Mexico State University, Las Cruces, NM, USA. ⁸Department of Family and Community Health, Fred N. Binka School of Public Health, University of Health and Allied Sciences, Hohoe, Ghana. ⁹School of Public Health, Bielefeld University, P.O. Box 100131, 33501 Bielefeld, Germany. ¹⁰Department of Industrial and Health Sciences, Faculty of Applied Sciences, Takoradi Technical University, P.O. Box 256, Takoradi, Ghana.¹¹Neurocognition and Action-Biomechanics-Research Group, Faculty of Psychology and Sport Sciences, Bielefeld University, Bielefeld, Germany.¹²School of Public Health, Faculty of Health, University of Technology Sydney, Sydney, Australia.

Received: 11 January 2022 Accepted: 21 May 2023 Published online: 29 June 2023

References

- Taye A, Asmare I. Prevalence of premarital sexual practice and associated factors among adolescents of Jimma Preparatory School Oromia Region, South West Ethiopia. J Nurs Care. 2016;5(2):353.
- Beyene K, Hinkosa L, Bekele D, Demtsu B. Magnitude of premarital sexual practice and associated factors among adigrat high school students, Ethiopia, a cross sectional study. EJRH. 2019;11(3):10.
- Morris JL, Rushwan H. Adolescent sexual and reproductive health: the global challenges. Int J Gynaecol. 2015;131:S40–2.
- Biro FM, Chan YM. Normal puberty. UpToDate Duryea TK, Snyder PJ, Geffner ME. Waltham: Upto Date. 2017.
- Hagisso SN, Handiso TB. Prevalence of premarital sexual practice and associated factors among Goba town high school students, South East-Ethiopia. Intensive Care Med. 2020;5:001–6.
- Shongwe MC, Chung MH, Chien LY, Chang PC. Does parenting style moderate the relationship between parent-youth sexual risk communication and premarital sexual debut among in-school youth in Eswatini? PLoS ONE. 2021;16(1): e0245590.
- Behulu GK, Anteneh KT, Aynalem GL. Premarital sexual intercourse and associated factors among adolescent students in Debre-Markos town secondary and preparatory schools, north west Ethiopia, 2017. BMC Res Notes. 2019;12(1):1–6.
- Berihun H. Assessment of the prevalence of premarital sex and unprotected sexual practices among secondary school adolescent students in Sebeta town, Oromia regional state, Ethiopia. Addis Ababa: Addis Ababa University; 2014.
- Mengistie Z, Wolie E, Abawa E, Ebre E, Adera A. Knowledge attitude and practice towards premarital sex and HIV/AIDS among Mizan-Tepi University students, south west Ethiopia. Sci J Public Health. 2015;3(4):592–9.
- 10. Ondenge K, Gvetadze R, Otieno G, Gust DA, McLellan-Lemal E. Factors associated with age of first sex among women screened for an

observational contraceptive vaginal ring study in Kisumu, Kenya, 2014. Afr J Reprod Health. 2021;25(1):101–13.

- Amo-Adjei J, Tuoyire DA. Timing of sexual debut among unmarried youths aged 15–24 years in sub-Saharan Africa. J Biosoc Sci. 2018;50(2):161–77.
- 12. Shrestha RB. Premarital sexual behaviour and its impact on health among adolescents. J Health Promot. 2019;7:43–52.
- 13. GSS GHS, ICF International. Ghana demographic and health survey 2014. Rockville, Maryland, USA; 2015.
- Alhassan N, Dodoo FNA. Predictors of primary and secondary sexual abstinence among never-married youth in urban poor Accra, Ghana. Reprod Health. 2020;17(1):1–13.
- Long-Middleton ER, Burke PJ, Cahill Lawrence CA, Blanchard LB, Amudala NH, Rankin SH. Understanding motivations for abstinence among adolescent young women: insights into effective sexual risk reduction strategies. J Pediatr Health Care. 2013;27(5):342–50. https://doi.org/10. 1016/j.pedhc.2012.02.010.
- Bogale A, Seme A. Premarital sexual practices and its predictors among in-school youths of shendi town, west Gojjam zone, North Western Ethiopia. Reprod Health. 2014;11(1):1–9.
- Endazenaw G, Abebe M. Assessment of premarital sexual practices and determinant factors among high school students in West Shoa Zone, Oromia Regional State, Ethiopia. Sci J Public Health. 2015;3(2):229–36.
- Addila AE, Abebe ND, Abebe W, Turuse EA. Premarital sexual practice and its predictors among preparatory school students living with and without parents in Hossana Town, southern Ethiopia. Science. 2020;8(3):63–71.
- Taiwo PA, Ola RM. Attitude of in-school adolescents toward premarital sex in Ibadan North Local Government Area. Int J Health Soc Inq. 2020;6(1):1–12.
- Regassa T, Chala D, Adeba E. Premarital sex in the last twelve months and its predictors among students of Wollega University, Ethiopia. Ethiop J Health Sci. 2016;26(4):351–8.
- 21. Teferra TB, Erena AN, Kebede A. Prevalence of premarital sexual practice and associated factors among undergraduate health science students of Madawalabu University, Bale Goba, South East Ethiopia: institution based cross sectional study. Pan Afr Med J. 2015;20(1):1–10.
- Asante KO, Nketiah-Amponsah E, Andoh-Arthur J, Boafo IM, Ampaw S. Correlates of early sexual debut among sexually active youth in Ghana. Int Q Community Health Educ. 2018;39(1):9–17. https://doi.org/10.1177/ 0272684X18811016.
- Bhatta DN, Koirala AK, Jha N. Adolescent students' attitude towards premarital sex and unwanted pregnancy. Health Renaiss. 2013;11(2):145–9.
- Beguy D, Mumah J, Wawire S, Muindi K, Gottschalk L, Kabiru CW. Status report on the sexual and reproductive health of adolescents living in urban slums in Kenya. 2013.
- Folayan MO, Adebajo S, Adeyemi A, Ogungbemi KM. Differences in sexual practices, sexual behavior and HIV risk profile between adolescents and young persons in rural and urban Nigeria. PLoS ONE. 2015;10(7): e0129106.
- Odimegwu C, Somefun OD, Chisumpa VH. Regional differences in positive sexual behaviour among youth in sub-Saharan Africa. J Biosoc Sci. 2019;51(2):254–72. https://doi.org/10.1017/S002193201800010X.
- Uchudi J, Magadi M, Mostazir M. A multilevel analysis of the determinants of high-risk sexual behaviour in sub-Saharan Africa. J Biosoc Sci. 2012;44(3):289–311.
- Akibu M, Gebresellasie F, Zekarias F, Tsegaye W. Premarital sexual practice and its predictors among university students: institution based cross sectional study. Pan Afr Med J. 2017;28(1):1–9.
- 29. Gbagbo FY, Gbagbo JA. Commercial sex work among university students: a case study of four public universities in Ghana. BMC Womens Health. 2021;21(1):1–3. https://doi.org/10.1186/s12905-021-01251-2.
- Corsi DJ, Neuman M, Finlay JE, Subramanian SV. Demographic and health surveys: a profile. Int J Epidemiol. 2012;41(6):1602–13.
- Aliaga A, Ruilin R. Cluster optimal sample size for demographic and health surveys. In: 7th international conference on teaching statistics— ICOTS, vol. 7. 2006. p. 2–7.
- Von Elm E, Altman DG, Egger M, Pocock SJ, Gøtzsche PC, et al. The strengthening the reporting of observational studies in epidemiology (STROBE) statement: guidelines for reporting observational studies. Int J Surg. 2014;12(12):1495–9.

- UNICEF. A league table for teenage births in rich nations. 2001. https:// www.unicef-irc.org/publications/328-a-league-table-of-teenage-birthsin-rich-nations.html. Accessed 21 Nov 2021.
- Sari UHP, Moedjiono AI, Bustan MN. Determinants of premarital sexual behavior of young men in Indonesia (IDHS data analysis 2017). East Afr Sch J Educ Humanit Lit. 2019;2(11):666–71.
- Sari UHP, Moedjiono AI, Bustan NM. Dating behavior and age at first time having premarital sexual intercourse on young men in Indonesia. Open Access Maced J Med Sci. 2020;8(T2):119–22.
- Puspasari P, Sukamdi S, Emilia O. The correlation of reproductive health information exposure from media on premarital sexual behaviour IDHS 2012 analysis. Berita Kedokteran Masyarakat. 2012;33(1):67–74.
- Ghaffari M, Gharghani ZG, Mehrabi Y, Ramezankhani A, Movahed M. Premarital sexual intercourse-related individual factors among Iranian adolescents: a qualitative study. Iran Red Crescent Med J. 2016;18(2): e21220.
- Atwood KA, Kennedy SB, Shamblen S, et al. Reducing sexual risk taking behaviors among adolescents who engage in transactional sex in postconflict Liberia. Vulnerable Child Youth Stud. 2012;7(1):55–65. https://doi. org/10.1080/17450128.2011.647773.
- Okigbo CC, McCarraher DR, Chen M, Pack A. Risk factors for transactional sex among young females in post-conflict Liberia. Afr J Reprod Health. 2014;18(3):133–41.
- Salih NA, Metaferia H, Reda AA, Biadgilign S. Premarital sexual activity among unmarried adolescents in northern Ethiopia: a cross-sectional study. Sex Reprod Healthc. 2015;6(1):9–13. https://doi.org/10.1016/j.srhc. 2014.06.004.
- Noroozi M, Taleghani F, Merghati-Khoei ES, Tavakoli M, Gholami A. Premarital sexual relationships: explanation of the actions and functions of family. Iran J Nurs Midwifery Res. 2014;19(4):424.
- Hindin J, Hindin MJ. Premarital romantic partnerships: attitudes and sexual experiences of youth in Delhi, India. Int Perspect Sex Reprod. 2009;35:97–104.
- Yan H, Chen W, Wu H, Bi Y, Zhang M, Li S, Braun KL. Multiple sex partner behavior in female undergraduate students in China: a multi-campus survey. BMC Public Health. 2009;9(1):1–11. https://doi.org/10.1186/ 1471-2458-9-305.
- AkterHossen M, Quddus AHG. Prevalence and determinants of premarital sex among university students of Bangladesh. Sex Cult. 2021;25:255–74. https://doi.org/10.1007/s12119-020-09768-8.
- Betzer F, Köhler S, Schlemm L. Sex work among students of higher education: a survey-based, cross-sectional study. Arch Sex Behav. 2015;44(3):525–8.
- Adams M, Oye J, Parker T. Sexuality of older adults and the internet: from sex education to cybersex. Sex Relatsh Ther. 2003;18(3):405–15.
- Fatusi AO, Blum RW. Predictors of early sexual initiation among a nationally representative sample of Nigerian adolescents. BMC Public Health. 2008;8(1):1–14. https://doi.org/10.1186/1471-2458-8-136.
- Chanakira E, O'Cathain A, Goyder EC, Freeman JV. Factors perceived to influence risky sexual behaviours among university students in the United Kingdom: a qualitative telephone interview study. BMC Public Health. 2014;14(1):1–7. https://doi.org/10.1186/1471-2458-14-1055.
- Mampane JN. Exploring the "Blesser and Blessee" phenomenon: young women, transactional sex, and HIV in rural South Africa. SAGE Open. 2018;8(4):2158244018806343. https://doi.org/10.1177/2158244018 806343.
- Zembe YZ, Townsend L, Thorson A, Ekström AM. "Money talks, bullshit walks" interrogating notions of consumption and survival sex among young women engaging in transactional sex in post-apartheid South Africa: a qualitative enquiry. Glob Health. 2013;9(1):1–16. https://doi.org/ 10.1186/1744-8603-9-28.
- Oljira L, Berhane Y, Worku A. Pre-marital sexual debut and its associated factors among in-school adolescents in eastern Ethiopia. BMC Public Health. 2012;12(1):1–5. https://doi.org/10.1186/1471-2458-12-375.
- Chiao C. Community context and the prevalence of premarital sex among young women in Kenya and the Philippines: trends and differences from 1993 to 2003. Health Place. 2010;16(3):512–22.
- Wright PJ. Mass media effects on youth sexual behavior assessing the claim for causality. Ann Int Commun Assoc. 2011;35(1):343–85. https:// doi.org/10.1080/23808985.2011.11679121.

- Kamaruddin K, Razalli R, Ariffin A. Prevalence of premarital sexual intercourse and its associated factors: a cross-sectional study among adolescents in Malaysian secondary school. Int J Acad Res Bus Soc Sci. 2018;8(6):1252–64.
- Gyan Bahadur BC, Basel PL. Premarital sex behaviors among college youths of Kathmandu, Nepal. Kathmandu Univ Med J. 2013;11(1):27–31. https://doi.org/10.3126/kumj.v1111.11019.
- Stoebenau K, Heise L, Wamoyi J, Bobrova N. Revisiting the understanding of "transactional sex" in sub-Saharan Africa: a review and synthesis of the literature. Soc Sci Med. 2016;168:186–97. https://doi.org/10.1016/j.socsc imed.2016.09.023.
- Mturi AJ, Moerane W. Premarital childbearing among adolescents in Lesotho. J S Afr Stud. 2001;27(2):259–75. https://doi.org/10.1080/03057 070120049967.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

